Author's response to reviews

Title: The use of complementary medicine by a self-selected community group of Australian women with polycystic ovary syndrome

Authors:

Susan Arentz Ms (s.arentz@uws.edu.au)
Caroline A Smith Professor (caroline.smith@uws.edu.au)
Jason A Abbott A/Professor (j.abbott@unsw.edu.au)
Alan Bensoussan Professor (a.bensoussan@uws.edu.au)

Version: 3 Date: 4 August 2014

Author's response to reviews:

Abstract resubmitted. Edits to background, methods, results and conclusion sections to more closely conform with 'instructions to authors' for research articles. Specifically the Background was adjusted with reference to the purpose of the study and the context of the study in relation to current knowledge. The method section was expanded, outlining statistical methods and elaborating on data collection methods. Results additional (brief) results were provided and the conclusion included a few words to better summarize the content of manuscript. The word count for the abstract is 349.